Author's response to reviews

Title: An evaluation of a morphine public health programme for cancer and AIDS pain relief in Sub-Saharan Africa

Authors:

Dorothy E Logie (DeLogie@aol.com)
Richard Harding (richard.harding@kcl.ac.uk)

Version: 4 Date: 20 July 2005

Author's response to reviews:

Thank you for reviewing and accepting this manuscript. I have made all the required formatting changes as detailed in your email and therefore the manuscript is considered by the authors as now ready for publication.